

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Vatter, Johannes

Working Paper Well-being in Germany: What explains the regional variation?

Diskussionsbeiträge, No. 50

Provided in Cooperation with: Research Center for Generational Contracts (FZG), University of Freiburg

Suggested Citation: Vatter, Johannes (2012) : Well-being in Germany: What explains the regional variation?, Diskussionsbeiträge, No. 50, Albert-Ludwigs-Universität Freiburg, Forschungszentrum Generationenverträge (FZG), Freiburg i. Br.

This Version is available at: https://hdl.handle.net/10419/55851

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



DISKUSSIONSBEITRÄGE DISCUSSION PAPERS

Well-Being in Germany: What Explains the Regional Variation?

Johannes Vatter

No. 50 - February 2012

der Albert-Ludwigs-Universität Freiburg



Well-Being in Germany: What Explains the Regional Variation?

Johannes Vatter*

February 24, 2012

Abstract

This paper examines regional differences in subjective well-being (SWB) in Germany. Inferential statistics indicate a diminishing but still significant gap between East and West Germany, but also differing levels of SWB within both parts of Germany. The observed regional pattern of life satisfaction reflects macroeconomic fundamentals, where labour market conditions play a dominant role. Differing levels of GDP and economic growth have contributed rather indirectly to well-being such that the period since the reunification can be considered as a period of joyless growth. Moreover, the effects of unemployment and income differ in size between regions in such a way that one can assume increasing marginal disutility of unemployment. In total, approximately half of "satisfaction gap" between East and West Germany can be attributed to differing macroeconomic conditions. In contrast, the comparably high levels of life satisfaction in Northern Germany are driven mostly by couples and go along with significantly higher fertility rates. Overall, I conclude that comparisons of SWB within a single country provide valid information.

JEL-Classification: R10, I31

Keywords: subjective well-being, regional disparities, unemployment, economic growth, fertility rate

^{*}Forschungszentrum Generationenvertärge (Research Center for Generational Contracts), Albert-Ludwigs-Universität Freiburg (Freiburg University), D-79085 Freiburg, Germany (Phone +49-761-203-9236. Corresponding author: E-mail address: johannes.vatter@vwl-uni-freiburg.de). The author is grateful for helpful comments from Stefan Moog, Daniel Ehing, Klaus Kaier and Matthias Weber. All errors remain my own.

1 Introduction

Happiness is not just a popular topic for front-page stories. The vast mountain of economic literature on human well-being that has grown during the last two decades (see e.g. Kahneman and Krueger (2006), Blanchflower and Oswald (2008) or Oswald and Wu (2010)) suggests that individuals' subjective well-being (subsequently referred to as SWB) provides credible information. Hence, happiness economics or – more precisely – evidence based on self-assessed life satisfaction is a promising approach to specify our understanding of utility, to update our welfare reporting standards and to complement the traditional framework of revealed preferences (see e.g. Ng (2003)).

Moreover, analyses of reported life satisfaction are also highly recognized due to an increasing skepticism about the hitherto promises of further economic growth. Basically, one can find two lines of critique with respect to GDP as a key welfare indicator and major policy objective. The first line of critique has been established by the influential remarks of Richard A. Easterlin on the relationship between growth and well-being. Easterlins' analyses suggest not just a small but no significant relation at all (see Easterlin (1974) and more recently Easterlin (2005)). However, for developed societies this seeming paradox of a joyless material growth can be widely explained by material repletion, increasing inequality and the human ability to quickly adapt to higher levels of consumption (see e.g. Lane (2000) or Clark et al. (2008)). A second line questions the compatibility of economic growth and ecological sustainability and goes back to the early 1970s and the Club of Rome. 1 Although this critique rather is concerned about the consequences of economic growth in the long run, it has also stressed the importance of the question whether the GDP is a valid measure for todays quality of life.

As a consequence of this research, countless private and more and more public initiatives are searching for new ways to measure welfare in a more precise and forward-looking way than it is done by the GDP.² But serving both kinds of requirements – measuring current well-being precisely and including aspects of sustainability – seems to be an unsolvable task as it would include substantial trade-offs within one key indicator. One important subordinated question is, however, whether data on SWB is really able to measure the current level of well-being of a nation (or region) validly. In fact, social scientists have gained more and more confidence in using and interpreting data on SWB. Indeed, some have come to the conclusion that these data could solve the problem of how to measure welfare itself instead of using complicated indicators (see Kahneman et al. (2004), Layard (2010) or Diener and Tov (2012)). Among several caveats of the idea of accounting social well-being by using subjective information, a

¹More recently the main focus of this debate has shifted from the finiteness of natural resources to the aspects of climate change and the global distribution of resources.

²Probably the most recognized attempt to complement the GDP has been undergone by the Commission on the Measurement of Economic Performance and Social Progress which was assigned by the French government. Additionally, the European Commission ("beyond gdp") and the OECD ("better life initiative") have become active in this field. And also in Germany an Enquete-Commission titled "Growth, Welfare, Quality of Life" ("Wachstum, Wohlstand, Lebensqualität") has recently been established in order to develop an alternative concept to measure the quality of life that enables politicians and the society as a whole to operate and decide on the basis of valid welfare indicators and without being short-sighted.

striking one has been the lack of international comparability. Whether an indicator is used or not, crucially depends on the question whether it is internationally or regionally comparable or not. If it is not certain that country A is really doing better than country B, it is rather useless to search for root causes of different levels of well-being which in turn could serve as an evidence base for policy measures. This issue becomes even more delicate if countries or political unions such as the EU explicitly try to balance unequal living conditions.

This paper discusses the applicability of self-assessed life satisfaction as an additional welfare indicator by focussing on regional differences within Germany. Thereby three major questions shall be answered: First, do regional mean values of reported life satisfaction reflect real differences in well-being? Second, by how much can the regional variation be explained by established welfare indicators such as GDP or the unemployment rate? Third, what else contributes to the observed regional pattern of SWB within Germany? Finally, I also shortly discuss the question whether SWB might be a proper instrument to complement the existing reporting systems and which determinants of national well-being should receive more attention.

The remainder of the article is as follows. Section 2 provides a short insight into related research. Section 3 discusses the database and provides some descriptive statistics. In Section 4, I show up to which point the regional variation of SWB can be explained by macroeconomic fundamentals such as GDP and unemployment. Section 5 provides some additional explanations for the comparably low levels of SWB in East Germany as well as for the 'nordic happiness puzzle'. Section 6 summarizes.

2 Related literature

Especially for developed countries the demand for adequate regional comparisons on welfare has significantly increased during the past years. However, some authors are skeptical with respect to the validity of international comparisons based on subjective information (e.g. Sachverständigenrat (2010)). The arguments are manifold: Simple cross-country comparisons might become misleading due to different regional perceptions of the question (e.g. Wierzbicka (2005)). It can also be argued that regional aspiration levels are differing such that it comes to an unequal interpretation of the given answering scale. Moreover, cultural differences and social norms can lead to special respond behaviors, and therefore, cause additional biases (e.g. Vittersø et al. (2005)).

While most articles do not thoroughly discuss these pitfalls, some authors have started to examine cross-country panel data and included country fixed-effects in order to avoid unobserved regional effects within cross-sectional observations (see e.g. DiTella et al. (2003), Helliwell (2003) or Alesina et al. (2004)). Cross-national surveys of this kind have helped to estimate the impact of different political constitutions, institutions or macroeconomic conditions on overall life satisfaction without denying differences in mentality and cross-sectional biases. Other authors like Kristensen and Johansson (2008) have stressed these issues even more and use vignettes to control for cultural biases to receive more valid, and therefore, comparable figures. Despite of such examples, a major expla-

nation for the relatively incautious use of international (or interregional) data might be its plausible pattern. Among several international rankings of wellbeing the most serious ones are based on the Eurobarometer, the World Value Survey, the World Gallup Poll and the International Social Survey Program. These international sources strongly suggests that reported life satisfaction is rather determined by objective conditions and obvious cultural achievements than by language, adopted aspiration levels or unobserved mentalities (see Diener and Klingelmann (2000), Veenhoven and Ehrhardt (1995), Diener et al. (1995), Frey and Stutzer (2002), and Sacks et al. (2010)). In fact, all data generally show significant correlations with objective life conditions such as health standards or income.

However, while traditional welfare indicators such as the GDP are able to explain large parts of the global variation in SWB, this is not necessarily the case for the developed world. As on the individual level (see Layard et al. (2008)), several authors stress that additional income is becoming also less important or even obsolete on an aggregated level (see e.g. Lane (2000), Oswald (1997) or Easterlin (2005)). More recent research suggests, however, that higher levels of economic output still raise reported well-being significantly even if many materialistic needs have been satisfied already (Clark et al. (2008), Blanchflower and Oswald (2004), and Shields and Price (2005)). In an ordered probit model using data from the Eurobarometer for 12 European countries, Di Tella et al. (2003) find significant effects for both the level as well as the fluctuation of GDP. Helliwell (2003) explores regional differences among 14 industrial countries by using the World Value Survey and finds decreasing but positive effects of aggregated income as well. Significant and notable effects of GDP per capita are also found by Stewart (2005). Sacks et al. (2010) confirm the general finding that marginal utility of GDP diminishes, however, they also argue that there are still notable effects even in the highly developed world. Overall, GDP is found to explain the international pattern of SWB (still) quite well, even within the developed world.

The probably second most recognized welfare indicator – the unemployment rate – and its impact on the overall life satisfaction of a society have been studied extensively as well. Without controversy are the strong and durable effects of unemployment on the individual level (see, e.g. Clark and Oswald (1994) or Winkelmann and Winkelmann (1995)). Aiming on the effects on the aggregated level, DiTella et al. (2001) estimate the social welfare tradeoff between unemployment and inflation using a cross-country panel. After controlling for several personal characteristics, country fixed effects, and year effects, they conclude that unemployment marginally leads to higher welfare losses compared to inflation. Unemployment is also studied by DiTella et al. (2003) for 12 European countries. Using data from 1975 to 1992, their results suggest that unemployment has been less important than GDP when it comes to the regional variation of SWB.

Harder to grasp but also relevant for overall levels of SWB seems to be the degree of inequality within one country. Oishi et al. (2011), Rousseau (2009) as well as Böhnke and Kohler (2010) argue that rising income inequality is a key factor to understanding the moderate increase in SWB within many developed countries. As a consequence, differing levels of inequality presumably also explain parts of the international variation in SWB. In comparison to other influential variables, however, the perception and impact of inequality probably vary considerably between nations (Alesina et al. (2004)). Furthermore, research on happiness and inequality is not that voluminous due to the lack of long and comparable cross-country panel data. Several authors also stress the relevance of different kinds of social capital. For example, Helliwell (2003) finds a significant relationship between the level of communal responsibility of a society and the level of life satisfaction. Bjørnskov (2006) emphasizes the different aspects of social capital and argues that indicators of social trust are strongly related to life satisfaction. Overall, it can be stated that an increasing and significant share of the international variation in SWB can be explained in a plausible way based on a manifold set of objective variables. The listed examples provide just a selection of a large number of articles published during the last years.

In contrast to international comparisons, regional analyses of SWB within one country have the advantage that all regions (in most cases) share one language and homogenous cultural traditions which reduces the potential of unobservable or just unobserved heterogeneity. On the other hand, differences in SWB obviously tend to be smaller and therefore more difficult to detect and to explain statistically. Recent studies, however, have shown for example significant effects of objective living conditions. Oswald and Wu (2010) provide evidence for the US in which SWB goes along significantly with a bundle of objective indicators for quality of life. Frey and Stutzer (2002) focus on the impact of institutional design with cross-regional data from Switzerland. Luechinger (2011) provides evidence on the relationship of air quality and life satisfaction with data from German Socio-oeconomic Panel (SOEP) and high-resolution SO₂ data.

With respect to Germany, however, most articles have studied regional differences on a descriptive level. Berlemann and Kemmesies (2004) reported regional aspects of well-being for Germany based on the Eurobaromenter. Maddison and Rehdanz (2007) argue that regional differences in SWB within Germany are rapidly converging due to migration and the adjustment of rent prices. Bergheim (2008) analyses SOEP data on a high-resolution level and emphasizes the correlation of fertility rates and life satisfaction. Easterlin and Plagnol (2008) focus on the evolution of SWB in East and West Germany after reunification. Raffelhüschen et al. (2011) compare the regional pattern of SWB with those of several well-explored drivers of well-being and find unexpectedly high satisfaction levels in northern regions. However, to my knowledge there has been no study that explicitly analyzes the drivers of regional variation in SWB within Germany.

3 Database

3.1 Datasource

The following analyses are based on the German Socio-oeconomic Panel (subsequently referred to as SOEP), a longitudinal survey that exists since 1984. The question on overall life satisfaction - "Taken all things together, how satisfied are you with life?" (Wie zufrieden sind sie, alles in allem, mit Ihrem Leben?) - is answered by choosing a level on an eleven-point-scale (ranging from "not satisfied at all" = 0 to "absolutely satisfied" = 10). Besides this information the panel contains a wide range of socio-economic variables on both individual and household characteristics. The SOEP database has started for West Germany with a sample of 12245 observations which has been to small to conduct extensive regional analyses. Subsequently however, the number of observations has increased due to added subsamples as well as data refreshments. With more than 24500, the largest number of observations was achieved in the year 2000. In the following years the number has slightly decreased according to panel mortality. In the wave of the year 2009, which has been the one recently available, 18602 interviews have been conducted. All respondents are of age 16 or older. The long time span of the panel, the considerable variety of socioeconomic variables and the large number of observations have made the SOEP to one of the most frequently used datasets for exploring the evolution and causes of subjective well-being (see Wagner et al. (2007)).

3.2 Regionalization

The SOEP is used for cross-regional analyses only rarely. This mainly has to do with the fact that the sample design is done in a way that the observations are - accordingly weighted - representative for whole Germany but not necessarily for smaller regions within Germany. Despite of the subsample for East Germany that has been designed and added in 1990 in order to allow also separated analyses for West and East, the possibility of drawing detailed comparisons between single regions is limited due to the decreasing number of observations per region. Nevertheless, the approximately 20.000 observations that have been made each year since 2000 are spread over all parts of Germany according to the density of the population. The data are collected in a way that the number of observations is proportional to the population of the federal states (Bundesländer) and it reflects rural areas and agglomerations representatively. As the addresses of the respondents have been selected through a random-route-model, it seems reasonable to test to which degree the SOEP provides information on differences and possible explanations for the variance in SWB on a regional level.

In this analyses three levels of regional subdivisions are used. For large-scale comparisons or regressions on representative subsamples Germany is divided into four large regions (north, west, east and south).³ To avoid any confusion I will refer to the former political division of Germany as East and West Ger-

 $^{^{3}}$ To get a detailed picture of the regional setting see Figure 8 in the appendix. The map illustrates both the large-scale and the detailed subdivision into 19 regions.

many and to the regional subdivisions as eastern and western parts of Germany. Table 1 shows some basic descriptive figures according to this devision.

region	shortcut	mean	std. dev.	obs.
North	Ν	6.880	1.870	3014
West	W	6.861	1.813	5873
East	E	6.875	1.868	4725
South	S	6.518	1.836	4441
Germany	GER	6.804	1.848	18053

Table 1: SOEP regional levels of overall life satisfaction 2009

Sources: Own calculations based on the SOEP. Notes: All data have been analytically weighted.

In a second subdivision of 19 regions I follow mainly the borders of the German federal states. The states with the largest population (Bayern, Baden-Württemberg, Nordrhein-Westfalen and Niedersachsen) are split up in more or less equally populated parts according to regional administration borders (Regierungsbezirksebene). Other federal states (Saarland and Bremen) are included in other regions due to critically low numbers of observations. In this framework I obtain an adequate number of observations for cross-regional regressions while representativeness remains high (see Table 2).

region	shortcut	mean	std. dev.	obs.
Baden	BA	6.898	1.710	524
Bayern	BY	6.893	1.972	1543
Berlin	BE	6.703	1.883	683
Brandenburg	BR	6.338	1.736	791
Franken	FR	6.870	1.864	1072
Hamburg	HH	7.355	1.610	269
Hessen	HE	6.859	1.937	1220
Mecklenburg-Vorpommern	MV	6.418	1.811	475
Niedersachsen/Nordsee	NN	6.990	1.842	869
Niedersachsen/Hannover	NH	6.799	1.824	887
Nordrhein/Köln	NK	6.924	1.737	844
Nordrhein/Düsseldorf	ND	6.797	1.747	1005
Rheinland-Pfalz/Saarland	RS	6.977	1.869	1024
Sachsen	SN	6.569	1.898	1371
Sachsen-Anhalt	ST	6.491	1.738	772
Schleswig-Holstein	SH	6.881	2.060	514
Thüringen	TH	6.345	1.808	824
Westfalen	WE	6.810	1.761	1780
Württemberg	WÜ	6.854	1.819	1586
Germany	GER	6.804	1.848	18053

Table 2: SOEP regional levels of overall life satisfaction in 2009

Sources: Own calculations based on the SOEP. Notes: All data have been analytically weighted.

The range within the means amounts to roughly 1.0 with Hamburg (7.36) at

the top and Brandenburg (6.34) at the bottom. The overall mean is 6.80. The standard deviations alternate around 1.8. The number of observations within a single region varies from 1780 (Westfalen) to 269 (Hamburg). The average number of observations per region is $950.^4$

Finally, an even more detailed division of 96 regions is used which follows an often used spatial planning category (Raumordnungsregion). Although this subsample includes a lot of noise due to low numbers of observation, it is still helpful for additional analyses that include more detailed regional information.

3.3 Representativeness of regional data

The SOEP is designed in a way that it contains a representative sample for both, East and West Germany. Also the large-scale division can be assumed to be highly representative. However, as noted above, the specific representativeness for smaller regions is not necessarily given. The limited numbers of observations suggest that at least some regional subsamples are not fully representative with respect to key drivers of well-being. Hence, any over- or underrepresentation of particular socio-economic groups can lead to biased results.



Figure 1: Representativeness of regional data in 2009

Sources: Own calculations based on average gross wages from the national accounts of the federal statistical office (VGR) and unemployment data from the German unemployment insurance (BA). Notes: All data from the SOEP have been analytically weighted. The unemployment ratios represent values from spring 2009 as SOEP data is collected mainly during the first half of the year. All unemployment ratios are expressed as share of the total labour force.

Figure 1 compares both income and unemployment data for all 19 regions with official data from the national accounts of the federal statistical office and the unemployment insurance. While relative positions with respect to gross wages per employee are fully in line with the official numbers, regional unemployment ratios based on the SOEP only fit roughly to institutional data. Especially

⁴At this point I apologize for having ignored the true origin of the people of the Oberpfalz who are rather committed to the south Bayern instead of Franken.

in several states of East Germany (BR, TH and ST) the unemployment rates calculated on the basis of the SOEP exceed official numbers significantly. In turn, the group of unemployed seems to be underrepresented in Hamburg. This divergence suggests that smaller regions are biased to some extent to the one or the other direction which has to be taken into account for the interpretation of the subsequent analyses.

3.4 Significance of regional differences in SWB

Simple t-tests show that deviations in SWB from the German average by more than 0.1 can be perceived as significant in most cases. In 2009, SWB is above-average in HH, NN and RS on a 1 percent significance level. NK and BY as well show above average values (on a 5 percent level). BA, SH, FR, HE and WU do not reach significant above-average levels but do exceed the average level as well with high probabilities. The values for WE, NH and ND are close to the German average. BE lies below average on a 10 percent level and all regions that have formerly belonged to the German Democratic Republic (BR, MV, SN, ST and TH) show levels of SWB significantly below the German average (see Figure 2).





Sources: Own calculations based on the SOEP. Notes: Red color indicates values below the German average.

Table 3 shows the results of a simple dummy regression to provide an overview about the total size of regional differences in SWB for the year 2009. All coefficient sizes indicate the difference in SWB compared to Brandenburg, the region with the lowest values. Although absolute levels have converged between East and West Germany since 2005, also 20 years after the reunification of Germany considerable differences in SWB do exist. It can be seen that especially northern regions report relative high satisfaction values. Hamburg is significantly ahead

and also the regional sample for NN shows a significant higher level of SWB compared to other regions of West Germany. The economically strong regions in southern Germany (BA, BY, FR and WU) however, do not differ significantly from other regions within West Germany. Within East Germany, Berlin and Sachsen report the highest levels of SWB. Two-sample t-tests for comparing the independent regional samples lead to corresponding results (see Table 14 in the appendix).

region	shortcut	coef.	std. dev.	t
Hamburg (N)	HH	1.02	0.161	6.31
Niedersachsen/Nordsee (N)	NN	0.65	0.130	4.98
Rheinland-Pfalz/Saarland (W)	RS	0.64	0.127	5.02
Nordrhein/Köln (W)	NK	0.59	0.133	4.41
Baden (S)	BA	0.56	0.140	3.98
Bayern (S)	BY	0.55	0.121	4.57
Schleswig-Holstein (N)	SH	0.54	0.167	3.24
Franken (S)	FR	0.53	0.127	4.22
Hessen (W)	HE	0.52	0.121	4.29
Württemberg (S)	WÜ	0.52	0.116	4.46
Westfalen (W)	WE	0.47	0.109	4.31
Niedersachsen/Hannover (N)	NH	0.46	0.129	3.58
Nordrhein/Düsseldorf (W)	ND	0.46	0.120	3.83
Berlin (E)	BE	0.36	0.135	2.71
Sachsen (E)	SN	0.23	0.131	1.77
Sachsen-Anhalt (E)	ST	0.15	0.132	1.15
Mecklenburg-Vorpommern (N)	MV	0.08	0.159	0.5
Thüringen (E)	TH	0.01	0.128	0.05

Table 3: OLS regression with regional dummies for 2009

Sources: Own calculations based on the SOEP. Notes: Dependent variable: subjective well-being. The baseline is allotted to Brandenburg (BR) with a value of zero.

3.5 (In)validity of regional well-being

One could argue that some fraction of the variation of SWB does not reflect real differences in life satisfaction. Overall, it can be found three major sources for invalidity: First, due to the limited number of observations per region it is obvious that some parts of the variation can be ascribed to random errors. A series of randomly drawn subsamples with 900 observations shows that approximately 40 percent of the total standard deviation in the sample with 19 regions is caused simply by statistical uncertainty. Second, as representativeness is not totally given this also causes distortions that reduce validity which is probably the case for Hamburg and partly for East Germany (see Section 3.3). Third, there might exist regionally differing social norms that lead to peculiarities within respond behavior which do not reflect real levels of life satisfaction. Imagine, for example, that it is more common to complain about something openly for people in region A than for those in region B. On the other hand, it might be culturally anchored to show modesty in region B, while communication is more direct in region A. Such differences in mentality can potentially influence survey data and are difficult to detect.⁵

4 Macroeconomic fundamentals

4.1 Correlations

Employment and income have been detected and described as positive drivers of well-being in many cases both on the individual (see, e.g. Winkelmann and Winkelmann (1995), Gerlach and Stephan (1996)) as well as on the aggregate level (see e.g. DiTella et al. (2001), DiTella et al. (2003), and Stewart (2005)). The evidence endorses the fact that indicators such as GDP or unemployment rate are the most frequently and prominently reported ones in many countries. This leads to two relevant questions: First, by how much can regional differences within Germany be explained by different levels of GDP and unemployment? Second, how important are both welfare indicators in relation to each other? These questions are discussed in the following paragraphs.

The descriptive statistics shown in Section 3 generally support the theory that aggregated levels of SWB can be explained largely by objective macroeconomic variables such as unemployment rates or mean income.⁶ Unsurprisingly, SWB is negatively correlated with the degree of unemployment and positively correlated with the economic performance of each region. During the period from 2000 to 2009, for the sample of 19 German regions, the correlation coefficient of GDP and SWB has been on average .51. In contrast, the correlation between regional SWB and unemployment (measured as ratio of unemployed persons to the total civil workforce) has been -.81 (see Table 4).⁷ Real growth rates however, turn out to be insignificantly negatively correlated with both SWB and GDP levels which indicates a slow economic convergence within Germany.

	GDP per capita	economic growth	unemployment
SWB	.51***	14	81***
GDP per capita	1	18*	57***
economic growth		1	.09

Table 4: Correlation coefficients for 19 German regions from 2000 to 2009

Sources: Own calculations based on the SOEP, Statistisches Bundesamt and Bundesagentur für Arbeit. Notes: *** significant at 0.1 percent level; ** at 1 percent level; * at 5 percent level.

⁵Theoretically, one could also argue that language is understood not in the same way even within one country due to regionally differing idioms. This would imply that the meaning of "life satisfaction" varies somewhat across regions. Although I do not provide any evidence, I think that this is not of great importance as the notion "life satisfaction" (Lebenszufriedenheit) has presumably the same meaning throughout Germany.

⁶I do not focus on inflation as inflation rates have been moderate in Germany since the mid-1990s. Furthermore, regional differences have been quite small due to common monetary policy and highly integrated markets for consumer goods.

⁷The fact that GDP and unemployment are correlated themselves significantly (-.57) obviously complicates the interpretation of these numbers. However, the problem of multicolinearity is considered and found to be no obstacle for identifying the major driver of regional well-being.

The correlations become even more significant if one focuses on the average income instead of GDP. Figure 3 shows exemplarily correlations between reported well-being and aggregated data taken directly from the SOEP as well as from the federal statistical office and the German unemployment insurance for 2009. Average monthly income appears to be highly correlated with SWB, where both correlation coefficients are larger than 0.8. For unemployment can be found a similar (but negative) level of correlation (-.85). Using official unemployment rates, the coefficient is a bit weaker with -0.64 which can be explained partly by the described lack of representativeness. With respect to cross-sectional data, it can be said that the correlations are mainly driven by the differences between East and West Germany. As one takes a closer look at West Germany, however, data show that correlations do not remain significant (see Figure 3).



Figure 3: Life satisfaction and macroeconomic key indicators 2009

Sources: SOEP, Statistisches Bundesamt and Bundesagentur für Arbeit, own calculations. Notes: All data from the SOEP have been analytically weighted. The unemployment ratios represent values from spring 2009 as SOEP data is collected mainly during the first half of the year. All unemployment ratios are expressed as share of the total labour force.

A similar picture arises from the analyses on the level of 96 regions (see Figure 4). The correlations are equally apparent even though correlation coefficients are lower which can be explained to a large extent by the lower number of observations per region.⁸ In fact, it is difficult to say whether the correlation is really weaker or not. In general, the level of the coefficients suggests that also at this regional level GDP and unemployment have the ability to indicate regional welfare.

Figure 4: SWB and macroeconomic key indicators for 96 regions 2009



Sources: Own calculations based on the SOEP, Statistisches Bundesamt and Bundesagentur für Arbeit, own calculations. Notes: All data have been analytically weighted.

4.2 Some arguments on causality

As macroeconomic indicators are collinear themselves and often also go along with other characteristics of the regional subsamples (e.g. wealth), such correlations can only provide a first impression. However, it seems highly implausible for the data that the results are driven by reverse causality or confounding factors. Thus, the possibility that subjective well-being drives GDP or unemployment is excluded in the following analyses even though this might play an important role on the individual level. I also exclude the possibility that other factors simultaneously change the macroeconomic variables and subjective well-being, as factors that would potentially qualify as confounding factors such as for example culture and institutions remain quite constant over the different regions and/or the given time period.

Commonly, the presented correlations are explained by a combination of individual and social effects. The individual effect is caused directly: a person looses her job, another receives a higher salary, etc. Social effects, however, are multilayer. They appear first of all at the social environment of the respective person, as unemployment or changes in household income also affect family members. Second, macroeconomic conditions cannot just be described as the sum of microeconomic happenings. They also might influence the mood of an entire society. If unemployment rises, this also might induce the fear of many

 $^{^{8}\}mbox{Notice}$ that the regional values of SWB can not be reasonably compared due to the low number of observations.

employees to become unemployed (see e.g. Clark et al. (2009) and Luechinger et al. (2010)) and higher contribution payments. Correspondingly, an increase in production not just leads to higher household incomes but probably also swells tax revenues and contributes to brighter expectations. Finally, there is no need for mentioning the well known impact of economic growth on the unemployment rate itself.⁹ When it comes to GDP, the relationship is even more complex. Although higher levels of GDP per head commonly go along with higher incomes and more consumption, aggregated numbers on income tend to be like a black box. The examples have already been described by many researchers: Just think about expenditures for national defense or intelligence - it is hard to grasp the direct link that should cause any increase in subjective wellbeing. Furthermore, GDP obviously compromises consumption expenditures and investments alike. Fluctuations of investment or export activities (which mostly drive business cycles) are not supposed to raise life satisfaction immediately. The effects of economic growth on average SWB can also be reduced if for example growth boosts particularly earnings at the top of the income distribution where people realize comparably low levels of marginal utility (Layard et al. (2008)). Focussing on positional concerns or crowding out of social interactions can lead to arguments where SWB can be lowered by economic growth (see e.g. Pugno (2009) or Kolmos and Salamon (2008)). Finally, any regional redistribution of public recourses - which is done intensively within Germany additionally reduces the potential impact of changes in GDP. Hence, there are several arguments that lead to the assumption that the effects of output per head are less significant than those of the labour market situation.

4.3 Macroeconomic estimations

The empirical strategy follows mainly the approach of Di Tella et al. (2001). In order to tackle both dimensions of causality, macro- and microeconomic regression models are used and finally combined by including macroeconomic variables to microeconomic fixed effects estimations. While all microeconomic data have its source in the SOEP, all macroeconomic data have been drawn from the federal statistical offices and the German unemployment insurance.

To receive elementary information about the overall effects of macroeconomic indicators on regional SWB, I started by conducting pure macroeconomic regressions where the dependent variable equals the weighted mean values of reported life satisfaction SBW_{it} of each region i in the year t:

$$SWB_{it} = \beta X_{it} + \epsilon_i + \lambda_t + \mu_{it} \tag{1}$$

Besides the explanatory marcoeconomic regressors X_{it} , the model includes regional fixed effects ϵ_i as well as year dummies λ_t . The error term is represented by μ_{it} . Table 5 shows the results from such a cross-regional OLS regressions for the time period from 1995 to 2009.¹⁰ Regressions in columns (1) and (2) use

⁹Note that Okun's law has changed significantly in Germany during the last decade. Due to labour market reforms and the demographic changes ahead, Germany has been able to sustain employment even with average growth rates of less than one percent.

¹⁰The time span for the panel analysis is restricted as reliable data on regional unemployment and output is available only since 1995.

the log of real GDP per capita as independent variables whereas regressions in columns (3) to (5) focus on economic growth (ΔGDP).

By including year dummies and regional fixed effects, the results in column (1) are most meaningful. While the effect of unemployment turns out to be highly significant, significance of GDP is only at a 10 percent level. The relative size of standardized beta-coefficients suggests that differences in unemployment do explain a significantly higher share of the regional variation in SWB. At the margin, one percentage point of unemployment reduces SWB by 0.033 while an increase in GDP per capita by one percent raises SWB by approximately 0.004.¹¹ As a consequence, this means that a steady reduction of the unemployment rate by only one percentage point would trade off an increase in GDP per capita of approximately 7.8 percent. In comparison to the effects measured by Di Tella et al. (2003), who estimate a trade-off relation of approximately 1 to 2.8 percent examining data from the Eurobarometer from 1975 to 1992, the German marginal rate of substitution seems very high.¹²

Table 5: Explaining life satisfaction through macroeconomic fundamentals for 19 German regions from 1995 to 2009

	(1)	(2)	(3)	(4)	(5)
unemployment	0333***	0531***	0369***	0210***	0186**
in percent	(.0070)	(.0032)	(.0071)	(.0061)	(.0059)
lnGDP	.4444*	.1133			
per capita	(.2527)	(.0598)			
$\Delta~{ m GDP}$			0043	.0026	.0043
per capita			(.0040)	(.0048)	(.0046)
$-\beta_u/\beta_{gdp}$	1.4	8.6	19.1	17.4	9.7
year dummies	yes	yes	yes	yes	no
reg. fixed eff.	yes	no	yes	no	yes
Obs.	285	285	285	285	285
R^2	.924	.714	.924	.775	.795
$adj.R^2$.914	.697	.913	.758	.777

Notes: OLS-Regression with standard errors in parentheses. Dependent variable: regional mean of SWB. *** significant at 0.1 percent level; ** at 1 percent level; * at 5 percent level. Regressions in columns (1), (3) and (5) include country specific time trends for Berlin (BE) and Hamburg (HH).

However, several arguments can be brought up to explain the moderate relevance of GDP. First, the panel analyses is based on data from a time period in which Germany has been significantly richer and more saturated than the countries of the European sample during the 1970s and 1980s. Second, as one focuses on the effects within one country, one has to take the regional redistribution scheme into account which e.g. transferred considerable shares

¹¹Since $(ln(1.01) - ln(1))0.4444 \simeq 0.0044$. In a similar regression for a sample of 66 nontransition countries based on the World Value Survey Sacks et al. (2010) find a significant effect of 0.004 for any increase of real GDP by one percent, which resembles the results of this paper. However, as they do not control for unemployment, estimates cannot be compared easily.

¹²Note that the analysis of Di Tella et al. (2003) is based on a life satisfaction question that offers only four different satisfaction levels for reply. Thus, coefficients cannot be compared easily.

of GDP from West to East Germany during the last two decades. Third, economic growth has not proportionally contributed to wage increases (especially for blue-collar jobs) during the considered time span (see e.g. Fuchs-Schündeln et al. (2010)). Thus, the estimated trade-off relation is not implausible on first sight. The finding that unemployment probably plays a dominant role compared to growth can be underlined by the recent increase in SWB within East Germany. Since 2005, the unemployment rate in East Germany has dropped rapidly from 18.7 to 13.0 percent in 2009 while reported life satisfaction has increased significantly from 6.3 to 6.5. In contrast, the evolution of SWB after the renunification has not reflected the rapid growth rates of real incomes.

The regression shown in column (2) of Table 5 excludes regional fixed effects and provides a picture that is even more explicit. While the negative effect of unemployment is becoming stronger, the impact of GDP is still significant on the 10 percent level but has only one forth in size compared to column (1). Although multicollinearity weakens the significance, this strongly suggests that the existing regional differences in SWB can be explained much better by unemployment than by data on aggregated income respectively GDP. The regression results in columns (3) to (5) basically point into the same direction. In each of the regressions unemployment is highly significant whereas no significant effect of real economic growth can be found.¹³

	(1)	(2)	(3)	(4)
unemployment	0437**	0447***	0412***	0412***
in percent	(.0136)	(0058)	(.0072)	(.0063)
GDP	.0082	.0100*		
per capita	(0263)	(.0041)		
lnGDP			1027	.3332**
per capita			(.5411)	(.1231)
year dummies	yes	yes	yes	yes
reg. fixed eff.	yes	no	yes	no
Obs.	192	192	192	192
R^2	.828	.447	.828	.447

Table 6: Fixed effects OLS regression for regional life satisfaction from 2001 and 2009

Notes: OLS-Regression with robust standard errors (in parentheses). Dependent variable: regional mean of SWB. *** significant at 0.1 percent level; ** at 1 percent level; * at 5 percent level.

Before one gives a broader interpretation of these results, it is advisable to recheck the findings by conducting regression analyses for the sample of 96 regions as well. Table 6 shows the regression results for a panel with only two waves, one in 2001 and the other one in 2009. This limited number of points in time is sufficient due to the high number of observations in each wave and allows us to avoid the problem of detecting effects of growth on SWB that are time-lagged. The year 2001 has been chosen as a starting point because of a

¹³As high growth rates do not instantly change the level of wages and salaries, I also run several regressions with lagged growth variables which, however, do not change the results substantially.

drastic increase in the number of observations in the SOEP data since 2000 and the availability of regional unemployment data. All aggregated data have been taken again from the federal statistical office and from the unemployment insurance. Regressions in columns (1) and (2) use nominal GDP per capita levels (where GDP has been multiplied by a factor of 1000) as explanatory variable whereas regressions in columns (3) and (4) use log values.¹⁴

The results are similar to the ones above even though the data contains additional noise due to the low numbers of observation per region. The negative effect of the unemployment rate is highly significant. Also the size of the effect of unemployment is robust at a level slightly above 0.04 which is close to the previous results. The GDP on the other hand does not enter the estimation in a significant way, at least if regional fixed effects are included. If regional effects are excluded, significance of both welfare indicators is increasing as one would expect. However, as the point estimates of GDP and lnGDP are far from robust, there is no sense in deriving any trade-off relation from these estimations.

To sum up, the macroeconomic estimates provide evidence for the assumption that the total effect of one percentage point change of unemployment lies between -0.3 and -0.5 depending on the regional division and whether regional fixed effects are included or not. If one follows the estimation of regression (1) in Table 5, this would mean that a regional difference in unemployment of 6 percentage points (the difference between West and East Germany in 2009) leads to a gap in SWB of approximately 0.2 on the 0-10 scale. This implies that approximately one half of the difference in SWB between East and West Germany can be attributed to different levels of unemployment. Assuming a convergence just in terms of GDP would result in a reduction of the "satisfaction gap" by approximately 25 percent. However, these estimates should just be classified as a first rough approximation as other explanatory collinear variables (e.g. wealth) are not included to the model. The fact that SWB – if one looks at smaller regions – does not reflect differences in economic growth from a period of eight years is stunning.

4.4 Microeconomic estimations

In order to separate the effects on the individual from those on the aggregated level and to control for multicolinearity I continue with several fixed effect regressions, aiming at individual life satisfaction. The OLS regression equation on a pure microeconomic level is given by

$$SWB_{jit} = \beta X_{jit} + \alpha_i + \epsilon_j + \lambda_t + \mu_{it},$$
⁽²⁾

where SWB_{jit} is the well-being level reported by individual i in country j in year t, α_i is the individual fixed effect of individual *i*, ϵ_j is a regional fixed effect and λ_t is a year effect. As there is no need for the availability of macroeconomic data, the time span of the analyzed panel is extended by three more waves

¹⁴Note that regional price levels in general change uniformly do to highly integrated markets. Hence, relative changes in nominal values of GDP also reflect relative changes in real values. However, when it comes to real estate markets and rents, regional price dynamics exist to some extent.

(1992 to 1994). The set of personal characteristics X_{jit} has been chosen according to general literature findings. The regression results are shown in Table 7 and confirm standard findings where unemployment causes a significant drop in SWB and personal labour income enhances SWB with decreasing marginal returns. If one controls for income variables, unemployment reduces SWB of a person by approximately 0.4 on the 0-10 scale (see column (3)). Without controlling for income, the size of the effect is rather -0.6 (see column (2)). As those estimates are fairly robust, this already suggests that the individual losses in life satisfaction do not account for the major part of the aggregated effects estimated in Section 4.3. This can be shown by a simple calculation: If the unemployment rate increases by one percent and the labour force represents 70 percent of the adult population, this implies that the effects on the individual level lower the overall mean of life satisfaction by only 0.0042 (= 0.01 * 0.7 * 0.6) which is less than 15 percent of the macroeconomic effect (see in comparison Table 5). Income on the other hand raises SWB by approximately 0.003 if real net income increases by one percent. Hence, if GDP growth is reflected in growth in income, an increase of real GDP per capita by 10 percent should cause a durable aggregated effect of approximately 0.03 which accounts for two thirds of the macroeconomic effect measured above. This is plausible as GDP growth also contributes to public sector growth. However, as our macroeconomic estimation is not very precise, it is not reasonable to note this relation as a robust finding. In addition, several authors argue that on an aggregated level positional concerns potentially disperse large parts of the individual effects (see Section 2). Although one can not draw any conclusion on this question, I am highly confident in concluding that the indirect effects of unemployment are of higher importance than the ones that lies within GDP growth. Moreover, the finding of Winkelmann and Winkelmann (1995), who argue that the individual costs of becoming unemployed sum up to high financial amounts, can be recorded.

Further interesting results are: (i) Divorced persons are happier than single persons who have never been married. (ii) Disposable income of household members is at least as important for subjective well-being as personal income. (iii) The subjective health status plays a dominant role among explanatory variables. (iv) Life satisfaction is found to be U-shaped with respect to age. (v) On average, marriage causes higher effects than cohabitation. (vi) Owning real estate is connected to higher levels of SWB.

4.5 Combining both datasets

After having shortly introduced the results of basic microeconomic regressions, I return to the aggregated level. The combined regression equation is given by

$$SWB_{jit} = \alpha_i + \beta_1 ln (GDP)_j + \beta_2 U_j + \sum_{k=3}^N \beta_k X_{kit} + \epsilon_j + \lambda_t + \mu_{it}$$
(3)

where as before α_i denotes the individual fixed effect of individual *i*, ϵ_j the regional fixed effect of region *j* and λ_t the year dummy in year *t*. In addition, aggregated data on GDP and unemployment (*U*) are included to the model. The

Independent Variable	(1)	(2)	(3)	(4)
age	0730***	0278***	0728***	0406***
-	(.0038)	(.0022)	(.0037)	(.0022)
age ²	.0475***	.0047*	.0539***	.0097***
2	(.0042)	(.0020)	(.0041)	(.0020)
SHS ^a "very good"	.7286***	.7421***	.7355***	.7356***
	(.0135)	(.0111)	(.0132)	(.0113)
SHS "good"	.4095***	.3990***	.4094***	.3961***
	(.0082)	(.0067)	(.0081)	(.0068)
SHS "not that good"	5125***	5622***	5200***	5552***
	(.0126)	(.0087)	(.0124)	(.0089)
SHS "bad"	-1.4316***	-1.642***	-1.4528***	-1.6235***
	(.0305)	(.0163)	(.0299)	(.0166)
married	.4796***	.4012***	.4423***	.4572***
	(.0181)	(0139)	(.0176)	(.0142)
with partner	.3329***	.3045***	.3306***	.3072***
	(.0135)	(0.108)	(.0133)	(.0110)
divorced	.2171***	.1673***	.2058***	.1922***
	(.0240)	(.0207)	(.0236)	(.0209)
unemployed	3879***	6038***	3771***	5597***
	(.0344)	(.0110)	(.0339)	(.0112)
$\ln Y_p$.1116***		.1601***	
personal net income	(.0094)		(.0087)	
$\ln Y_h$.2193***			.2973***
hh net income per member	(.0123)			(.0089)
house/flat owner	.0938***	.0808***	.0949***	.0768***
	(.0121)	(.0100)	(.0119)	(.0102)
R-sq (within)	.0794	.0985	.0782	.1015
R-sq (between)	.1404	.1579	.1403	.1549
R-sq (overall)	.1336	.1433	.1344	.1383
rho	.5626	.5536	.5613	.5627
Number of individuals	30320	44222	31227	42938
Avg obs per ind.	5.9	7.3	6.0	7.3

Table 7: Fixed effects OLS regression for individual life satisfaction from 1992 to 2009

Notes: OLS-Regression with individual fixed effects, individual characteristics as well as aggregate information. Number of observations: 188069. Standard errors in parentheses. Dependent variable: regional mean of SWB. *** significant at 0.1 percent level; ** at 1 percent level; * at 5 percent level. Regional fixed effects and year dummies have been included. ^{*a*}SHS: subjective health status.

baseline results are shown in column (1) of Table 8. If one controls for all individual characteristics from regression (4) in Table 7, this yields in harsh drop of the effect of the unemployment rate by approximately one half. The effect, however, remains statistically significant on a 0.1 percent level which suggests that within the pure macroeconomic model unemployment rates serve also as proxies for other important variables such as mean household income, financial wealth or even average health status. Nevertheless, there remains a considerable macroeconomic effect of approximately -0.15 (per percentage point of unemployment). In contrast, GDP does not contribute to the estimation in a significant way. In the baseline regression (column (1)) the point estimate even turns out to be negative. This result does not even change qualitatively if regional fixed effects are excluded (see column (2)) which leads to the conclusion that the sum of all indirect effects of GDP growth have not contributed to regional well-being during the last 15 years. This, however, does not imply that GDP growth is obsolete for the country as a whole, although the negative coefficient for log GDP even turns out to be significant as soon as time and regional dummies are excluded (see column (4)). That this is a false conclusion can be easily seen if one considers the regional distribution of public net investments within Germany since the reunification. In fact, the stock of public infrastructure has increased in per capita terms especially in those regions where it has been on the lowest levels. This hypothesis is one out of many and cannot be discussed here in detail.

	(1)	(2)	(3)	(4)	(5)
unemployment (in percent) unemployment ²	0154*** (.0035)	0150*** (0029)	0307*** (.0020)	0287*** (.0019)	.0099 (.0092) 0008** (.0003)
lnGDP	2217	0524	1071	1286*	2420*
	(.1149)	(.0638)	(.1067)	(.0595)	(.1151)
year dummies	yes	yes	no	no	yes
reg. fixed eff.	yes	no	yes	no	yes
R-sq (overall)	.1375	.1279	.1132	.1172	.1421
rho	.5724	.5804	.5953	.5917	.5688

Table 8: Fixed effects OLS regression for individual life satisfaction with macroeconomic variables from 1995 to 2009

Notes: OLS-Regression with individual fixed effects and individual characteristics (see column (4) of Table 7). Standard errors in parentheses. Dependent variable: individual SWB.Number of observations: 276191. Number of individuals: 40906. Average number of observations per individual: 6.8. *** significant at 0.1 percent level; ** at 1 percent level; * at 5 percent level.

What can be discussed instead is how much of the regional variance can be explained by unemployment and differences in income. Figure 5 shows the unexplained regional effects of both the microeconomic as well as the combined regression model. In comparison to the total variation of SWB in 2009, which is illustrated by the dark bars, it can be seen that significant parts but not the majority of regional differences can be attributed to macroeconomic fundamentals. For most regions within East Germany, however, considerable negative effects still remain. In contrast, Hamburg and the very northwest of Germany still show surprisingly high values of SWB. Nevertheless, approximately 40 percent of the gap in life satisfaction between East and West Germany can be attributed to fundamental macroeconomic conditions.



Figure 5: Unexplained regional deviation 2009

Source: Own calculations based on the regression results in Tables 7 and 8.

4.6 The marginal disutility of unemployment

To complete the analyses of Sections 4.1 to 4.5, it can be argued that the impact of macroeconomic fundamentals is not equal in size within each region partly due to nonlinearity of effects. With respect to GDP and household income this has been considered by using log values as marginal utility diminishes both on the individual and on the aggregate level (Sacks et al. (2010)). However, when it comes to unemployment, things are less obvious. In fact, there are several arguments that counter the simple idea of constant marginal effects of unemployment as well. However, as one has to take into account always both groups, the employed and the unemployed, this is more tricky. In theory, the employed are harmed by unemployment at least in two ways: First, high unemployment is connected to lower job security. In other words, the higher the unemployment is the more afraid of layoffs employees become. Second, employees who witness layoffs might suffer as well due to feelings of guilt and a worsening of social climate at work. Some more linkages are discussed by Clark et al. (2009). What I want to emphasize is that these arguments only apply if unemployment exceeds some basic level. If, in contrast, unemployment is only a problem of persons with deficits in basic qualifications, it is not plausible that this causes any fear among the employed. The same is true if unemployment mainly occurs due to persons who enter the labour market after education or change jobs voluntarily. As a consequence, the negative effects of unemployment on the employed would only be considerable if unemployment is caused by structural labour market distortions such as high labour costs or business cycles which lead to unemployment rates that exceed a basic level. This can be a potential explanation for an increase in marginal disutility of unemployment on the aggregate level which is indicated by regression (5) in Table 8.

With respect to the unemployed Clark (2003) provides evidence from the British Household Survey and argues that higher unemployment numbers help the unemployed to cope with their "violation" of the social norm not to live from public funds. Hence, some authors have concluded that a rise in unemployment eases the disutility of the unemployed which as a consequence would contribute to a decrease in marginal disutility of unemployment. For Germany, however, empirical evidence is not in line with any conclusion that ascribes a dominant role to this effect. Regionally separated fixed effects regressions show significantly stronger effects of unemployment for eastern parts of Germany where unemployment rates have been nearly twice as high during the last two decades (see Table 9). On the other hand, the effect has been significant but rather low for southern Germany where unemployment has remained on a comparably low level during the same time period. Similar findings have been also made by Chadi (2011) who emphasizes the effect of lower job prospects in presence of high unemployment rates. Overall, I do not want to conclude on possible causes at this point, but there is evidence suggesting increasing marginal disutility for both groups, the employed as well as the unemployed. Hence, it can be assumed that differing levels of unemployment contribute even more to the regional variation of SWB, in particular with respect to the difference between East and West Germany.

5 Other explanatory factors

Obviously, the regional levels of well-being are not just a result from macroeconomic conditions. It can rather be assumed that numerous factors exert an influence on the overall SWB within one country or region while not all of them are easy to detect or durable over time. The following paragraphs discuss only two aspects that help understand the remaining pattern of unexplained regional SWB.

5.1 Communistic heritage

Thus far, the comparably low levels of SWB in East Germany have been explained to a large extent by lower incomes and higher unemployment rates. However, a considerable part of the satisfaction gap between East and West Germany remains unexplained. To solve this puzzle it is helpful to focus on single generations. Table 10 reports SWB for different age groups in West and East Germany for the year 2009. The values indicate strong cohort effects. Note that those aged 30 and older in 1990 report far lower satisfaction levels than their western counterparts. In comparison to the younger generation the difference in SWB is nearly twice as large for individuals born before 1960. In 2009, the difference in SWB between the generation aged 30 and younger of East and

Independent Variable	(north)	(west)	(south)	(east)
age	1136***	1156***	1287***	0451***
-	(.0030)	(.0189)	(.0205)	(.0106)
age ²	.0535***	.0479***	.0469***	.0463***
-	(.0136)	(.0097)	(.01060)	(.0120)
SHS ^a "very good"	.7585***	.7071***	.7662***	.6908***
	(.0369)	(.0259)	(.0272)	(.0321)
SHS "good"	.4078***	.4024***	.4258***	.3984***
	(.0231)	(.0155)	(.0178)	(.0187)
SHS "not that good"	4910***	5106***	5911***	4739***
	(.0396)	(.0267)	(.0309)	(.0327)
SHS "bad"	-1.4902***	-1.482***	-1.496***	-1.3141***
	(.0396)	(.0869)	(.0909)	(.0977)
married	.5253***	.4552***	.4825***	.3709***
	(.0626)	(0446)	(.0469)	(.0556)
with partner	.4034***	.3331***	.3227***	.3006***
	(.0433)	(.0313)	(.0326)	(.0370)
divorced	.1446***	.2567***	.2142**	.2177**
	(.0806)	(.0577)	(.0653)	(.0682)
unemployed	3973***	4803***	2093	5248***
	(.1150)	(.0866)	(.1094)	(.0700)
net labor income	.0092***	.0128***	.0113***	.0274***
in 100 Euro	(.0024)	(.0013)	(.0014)	(.0024)
net labor income ²	0012***	0015***	0014***	0058***
	(.0003)	(.0002)	(.0002)	(.0010)
house/flat owner	.0504***	.1008***	.07847*	.0317*
	(.0149)	(.0262)	(.0282)	(.0317)
R-sq (within)	.0794	.0865	.0898	.0669
R-sq (between)	.0346	.0302	.0329	.0019
R-sq (overall)	.0447	.0388	.0487	.0090
rho	.6668	.6631	.6950	.6842
number of obs.	29447	62240	51570	44812

Table 9: Fixed effects OLS regression for individual life satisfaction by region from 1992 to 2009

Notes: OLS-Regression with individual fixed effects, year dummies and regional fixed effects. Standard errors in parentheses. Dependent variable: regional mean of SWB. *** significant at 0.1 percent level; ** at 1 percent level; * at 5 percent level. Regional fixed effects and year dummies have been included. ^{*a*}SHS: subjective health status.

West Germany has been 2.5 times smaller compared to the difference between those aged 50 and older (see Table 10).

Table 10: SWB in East and West Germany for different age groups in 2009

age group in yrs.	20-29	30-39	40-49	50-59	60-69	70-79
West Germany	7.07	7.11	6.66	6.58	6.98	6.94
East Germany	6.89	6.81	6.39	6.07	6.50	6.46
Absolute difference	0.18	0.30	0.27	0.51	0.48	0.48

Source: SOEP. Notes: Weighted. Number of observations: 18053.

To complement this cross-sectional data, Figure 6 shows by how much the mean SWB of the respectively younger generation (35 years and below) exceeds the satisfaction level of the older generation (older than 35). While the relative happiness of the younger generation has only slightly increased in West Germany, it has risen dramatically for East Germany especially between 1995 and 2005: Directly after reunification there has been only a difference of approximately 0.2. In 2009, when the young generation consisted of individuals born after 1974, the difference in SWB between the generations accounted for more than 0.5.





Source: SOEP. Notes: Weighted. At any time the young generation consists of those younger than 35. The old generation is defined respectively as the group aged 35 and older.

These data suggest that those generations who have not just grown up within the German Democratic Republic, but also spend a considerable share of their working life under the communistic regime are significantly worse off. Whether this finding is connected mainly to difficulties of adjusting to the mechanisms of a free and competitive labour market, to the anger of having spend irretrievable years under limited freedom or to durable disappointments about the failure of socialistic policy – to name just a few possible root causes – cannot be discussed here. Nevertheless, it is safe to say that the low levels of reported life satisfaction in East Germany are still linked to the cultural imprints of many who have spent parts of their lives in the former DDR. Thus, it can be assumed that one would perceive a partial convergence of SWB in the future even if macroeconomic conditions do not continue to converge.

5.2 A note on the nordic happiness puzzle

Besides the huge differences between East and West Germany, there are also comparably high values of reported life satisfaction in northern Germany which remain mostly unexplained (see Figure 5). Again, it is helpful to analyze the data separately. Table 11 shows mean values of SWB of different age groups for each part of Germany. Note that the relatively high levels of life satisfaction within the North of Germany are mainly driven by the elderly. For the group aged 60 and older, northern Germany reports considerably higher numbers (7.08) than their economically stronger counterparts in southern Germany (6.90), whereas the generation between 35 and 60 reports lower levels of SWB than this generation in the West and the South of Germany.

Table 11: Regional SWB by age groups in 2009

	North	West	South	East
overall	6.88	6.86	6.88	6.52
under 35 yrs.	7.12	7.03	7.10	6.99
35 to 60 yrs.	6.66	6.76	6.75	6.28
over 60 yrs.	7.08	6.92	6.90	6.48

Source: SOEP. Notes: Weighted. Number of observations: 18053.

These high values of SWB can neither be explained by differing levels of unemployment nor by differing income structures as both variables would indicate an inverse relation. One indication for the origin of this phenomenon can be found within family life. Table 12 summarizes the satisfaction responses for all persons aged 60 or older by marital status. While the share of people without partner is fairly equal in each region, reported life satisfaction differs strongly between both regions and according to marital status. As one would expect, singles report the lowest levels of life satisfaction. Probably more surprising is the fact, that cohabiting individuals of this generation report higher values of SWB than married ones. Most notable however is the large discrepancy between singles and those with partner in northern Germany. In fact, the phenomenon of "nordic happiness" seems to be driven mostly by persons aged 60 and older, who are married or live in cohabitation. With means of 7.2 and 7.4, the reported life satisfaction of these subgroups is significantly higher compared to western or southern regions of Germany which is not the case for single persons.

These descriptive findings are also in line with the results from the regional regression analyses. Separate fixed effect estimations for each part of Germany show significant differences in effect sizes for marriage and cohabitation (see Table 9). While regression coefficients for marriage and cohabitation are lowest in eastern Germany, the largest effects are found for northern Germany. However, if the exceptional levels of SWB in some parts of northern Germany are

Table 12: Regional SWB by marital status in 2009

Region	North	West	South	East
married partner	7.20 7.38 6.70	6.99 7.12 6.77	6.97 7.13	6.42 7.23
Siligic	0.70	0.77	0.00	0.1

Source: SOEP. Notes: Weighted. Number of observations: 5810.

partly due to the well-being of couples, the question arises whether there exist objective indicators which support this argument. Table 7 presents data on regional fertility related to SWB (left graph) and GDP (right graph). Fertility is indicated here by the number of children younger than 6 related to the total population.

Figure 7: SWB and relative number of children for 96 regions 2009



Source: SOEP, Statistisches Bundesamt, and Bundesamt für Bauwesen und Raumordnung. Notes: Weighted. All observations from north-western regions (HH, NN and SH) which report comparably high levels of SWB are marked black.

What can be seen is that most of the regions in north-western Germany do show moderate levels of GDP, but originate comparably high numbers of children. The significant correlation between large numbers of children and life satisfaction can be interpreted in several ways. For Germany, most studies do not show strong effects on parental life satisfaction (see recent work by Humpert (2010) and Raffelhüschen et al. (2011)). On the other hand, it is highly plausible that happy couples tend to have more children. Hence, the number of children probably serves well as an objective indicator but not necessarily as a source of the happiness of nordic couples. This, however, does not perfectly fit the descriptive results which emphasize the role of the older generation (see Table 11). In fact, most studies that examine the effect of children on parental life satisfaction focus on newborns and younger children due to limited data. This often leads to the trivial conclusion that children are a source of both joy as well as time pressure and stress which is why many studies do not find significant effects or even negative ones (see e.g. Clark and Oswald (1994) or Raffelhüschen et al. (2011)). However, there is only limited evidence on the overall effects of grown-up children and grandchildren. Table 13 shows the results of a simple cross-sectional regression for 96 regions. After controlling for GDP and unemployment, the relative number of children younger than 6 within a region still remains positively correlated with SWB. This positive relation between regional SWB and regional fertility in combination with the outstanding values of well-being reported by elderly couples in regions with high fertility rates supports the theory that children do contribute to higher levels of SWB, at least as soon as they have flown out to have their own children.¹⁵

variable	coef.	std. error	t	beta
lnGDP	.3556*	.1839	1.93	.1949
unemployment rate	0214*	.0118	-1.82	1938
share of children < 6	.2189*	.1239	1.75	.2255

Table 13: OLS-Regression for 2009

Notes: OLS-regression with robust standard errors. Dependent variable: regional mean of SWB. Number of observations: 96, $R^2 = 0.27$, *significant on a 10 percent level.

6 Summary and discussion

This paper examines regional differences in SWB within Germany. Although the analyses do only cover some out of many possible drivers of SWB, I conclude that large parts of the variation do reflect objective living conditions. Moreover, highly recognized welfare indicators such as the GDP or unemployment rates are not just correlated with regional SWB but also have the ability to explain the evolution and differences of regional levels of SWB.

The macroeconomic regression model applied for a sample with 19 regions with mean values of SWB as dependent variable suggests for the period from 1995 until 2009 that up to three-quarters of all real differences are related to macroeconomic conditions. If common statistical standard errors are taken into account, still one half of the total variation can be attributed to macroeconomic conditions. These estimates, however, include presumably various indirect effects which go along with fluctuations of unemployment or growth. To obtain more precise estimates several microeconomic regressions are conducted as well. Applying common FE-regressions, it can be found that that regional differences in household income and individual effects of unemployment do explain approximately 25 percent of all real variation. If the macroeconomic effects from unemployment are taken into account as well, this fraction increases to approximately 40 percent which comes close to the results from our rather simplistic macroeconomic model. In addition, I conclude that there is strong evidence that marginal disutility is increasing with unemployment. As a consequence, it can be assumed that the differences in quality of life between East

¹⁵Unfortunately, the SOEP (like most panel data) does not contain information on the number of grandchildren a person has.

and West Germany would be reduced by approximately half if macroeconomic conditions were equal.

The main question on the relative importance of unemployment and growth can be answered unambiguously. Even though both welfare indicators are highly correlated with SWB, the results strongly suggest that differing levels of unemployment do affect regional SWB significantly stronger than differences in GDP. Correspondingly, also changes in unemployment influence SWB stronger and more directly than economic fluctuations. Correspondingly, there has been no significant relation between economic growth rates and well-being for the analyzed period of time. Controlling for income effects on the household level, the model even yields negative though insignificant coefficients for GDP. However, this does not necessarily mean that economic growth can be fully neglected. Beyond dispute are the indirect effects of growth as it maintains and creates employment. Moreover, our analyses is short-sighted with respect to regional transfers within the public budget. Finally, the moderate impact of growth can be explained by the sharp decline of the labour share and the corresponding increase in inequality in Germany since the early 1990s. Hence, the interdependency probably will change again if Germany experiences higher wage increases in the future.

Although labour market conditions and economic growth still play an important role for overall life satisfaction, regional SWB cannot be explained fully by these factors. Approximately one half of the total variation of SWB remains unexplained. Especially the large gap between West and East Germany as well as exceptionally high levels in northern Germany are nebulous. In this regard, it can be argued that considerable cohort effects still delay higher levels of SWB in East Germany where people born before 1965 report significantly lower levels of life satisfaction while members of younger generations shows trends of steady convergence. Finally, I provide one piece of information on the 'nordic happiness puzzle'. As high levels of SWB are mostly driven by couples, the analysis focusses on family life. One objective indicator that is in line with this regional phenomenon is the fertility rate which is significantly above average in northwestern parts of Germany. In view of recent studies this yields two hypotheses which underline the validity of northern happiness: First, happy couples tend to have more children and second, grandchildren increase life satisfaction. However, I also perceive a need for additional studies on regional mentalities, which cannot be excluded as potential drivers of regional SWB.

When it comes to the discussion on new sets of welfare indicators, SWB definitely provides an important basis to detect relevant drivers of welfare as it contains valid information. Hence, it seems to be advisable for governments to collect more data on aspects that presumably influence the well-being of society. For example, it is highly probable that reliable data on regional income inequality would lead to several important and influential studies. This, in turn, can help to design indicators for those characteristics which are known for affecting SWB. Moreover, I do not perceive any fundamental caveat for using data on SWB in order to measure welfare directly, at least within one culturally and linguistically homogenous region. To reduce statistical uncertainty, however, it would be helpful to include subjective information of this kind into larger cross-sectional surveys such as common census data.

A Appendix



Figure 8: Regional subdivisions

	BA	BY	BE	BR	FR	HH	HE	MV	NN	NH
BA		52.1	96.8	100.0	61.3	0.0	65.2	100.0	17.6	84.1
BY	47.9		98.3	100.0	61.6	0.0	67.0	100.0	11.6	87.5
BE	3.2	1.7		100.0	3.4	0.0	4.4	99.5	0.1	15.3
BR	0.0	0.0	0.0		0.0	0.0	0.0	21.7	0.0	0.0
FR	38.7	38.4	96.6	100.0		0.0	55.0	100.0	7.8	80.0
HH	100.0	100.0	100.0	100.0	100.0		100.0	100.0	99.8	100.0
HE	34.8	33.0	95.6	100.0	45.0	0.0		100.0	6.1	76.4
MV	0.0	0.0	0.5	78.3	0.0	0.0	0.0		0.0	0.0
NN	82.4	88.4	99,9	100.0	92.2	0.2	93.9	100.0		98.5
NH	15.9	12.5	84.7	100.0	20.0	0.0	23.6	100.0	1.5	
NK	60.7	65.0	99.1	100.0	74.1	0.0	78.0	100.0	22.2	92.7
ND	14.1	10.6	85.4	100.0	18.0	0.0	21.5	100.0	1.0	48.9
RS	79.3	86.2	99.8	100.0	90.6	0.1	92.8	100.0	44.2	98.2
SN	0.0	0.0	6.6	99.7	0.0	0.0	0.0	93.5	0.0	0.2
ST	0.0	0.0	1.3	95.9	0.0	0.0	0.0	76.0	0.0	0.0
SH	44.3	45.4	94.0	100.0	58.1	0.0	58.1	100.0	15.4	77.9
TH	0.0	0.0	0.0	53.1	0.0	0.0	0.0	24.2	0.0	0.0
WE	15.6	10.1	90.7	100.0	23.3	0.0	23.3	100.0	0.7	55.6
WÜ	31.4	28.5	96.4	100.0	41.4	0.0	46.9	100.0	3.9	76.3
GER	12.5	3.6	91.9	100.0	12.8	0.0	15.4	100.0	0.2	52.7
	NK	ND	RS	SN	ST	SH	TH	WE	WÜ	GER
BA	39.3	85.9	20.7	100.0	100.0	55.7	100.0	84.4	68.6	87.5
BY	35.0	89.4	13.8	100.0	100.0	54.6	100.0	89.9	71.5	96.4
BE	0.9	14.6	0.2	93.4	98.7	6.0	100.0	9.3	3.6	8.1
BR	0.0	0.0	0.0	0.3	4.1	0.0	46.9	0.0	0.0	0.0
FR	25.9	82.0	9.4	100.0	100.0	45.8	100.0	80.6	58.6	87.2
HH	100.0	100.0	99.9	100.0	100.0	100.0	100.0	100.0	100.0	100.0
HE	22.0	78.5	7.2	100.0	100.0	41.9	100.0	76.7	53.1	84.6
MV	0.0	0.0	0.0	6.5	24.0	0.0	75.8	0.0	0.0	0.0
NN	77.8	99.0	55.8	100.0	100.0	84.6	100.0	99.3	96.1	99.8
NH	7.3	51.1	1.8	99.8	100.0	22.1	100.0	44.4	23.7	47.3
NK		94.0	26.2	100.0	100.0	66.0	100.0	94.0	82.0	96.8
ND	6.0		1.3	99.9	100.0	20.4	100.0	42.8	21.6	45.7
RS	73.8	98.7		100.0	100.0	82.2	100.0	99.1	95.3	99.8
SN	0.0	0.1	0.0		82.8	0.1	99.7	0.0	0.0	0.0
ST	0.0	0.0	0.0	17.2		0.0	94.9	0.0	0.0	0.0
SH	34.0	79.6	17.8	99.9	100.0		100.0	78.0	61.0	82.4
ΤН	~ ~	0.0	0.0	0.2	51	0.0		0.0	0.0	0.0
	0.0	0.0	0.0	0.5	5.1	0.0		0.0	0.0	0.0
WE	0.0 6.0	57.2	0.0	0.3 100.0	100.0	22.0	100.0	010	23.7	55.3
WE WÜ	0.0 6.0 18.0	57.2 78.4	0.0 0.9 4.7	0.3 100.0 100.0	100.0 100.0	22.0 49.0	100.0 100.0	76.3	23.7	55.3 85.1

Table 14: Probabilities of higher regional SWB in percent, 2009

Source: Own calculations based on the GSOEP. Notes: Two-sample t-tests. Values denote probabilities by which the SWB of a region in a certain row is higher than the one in the region of the respective column.

References

- Alesina, A., R. DiTella, and R. J. MacCulloch (2004). Inequality and happiness: are europeans and americans different? *Journal of Public Economics* 88(9-10), 2009–2042.
- Bergheim, S. (2008). Well-being in germany its happy regions have much in common. *Deutsche Bank Research*.
- Berlemann, M. and C. Kemmesies (2004). Zur entwicklung der lebenszufriedenheit nach der deutschen wiedervereinigung - eine empirische analyse in sachsen, ost- und westdeutschland. *ifo Dresden berichtet* (6), 3–10.
- Bjørnskov, C. (2006). The multiple facets of social capital. *European Journal of Political Economy 22*(1), 22–40.
- Blanchflower, D. and A. J. Oswald (2008). Hypertension and happiness across nations. *Journal of Health Economics* 27(2), 218–233.
- Blanchflower, D. G. and A. J. Oswald (2004). Well-being over time in britain and the usa. *Journal of Public Economics* 88(7-8), 1359–1386.
- Böhnke, P. and U. Kohler (2010). Handbook of European Societies. Social Transformations in the 21st Century, Chapter Well-being and Inequality, pp. 629– 666. New York: Springer Science.
- Chadi, A. (2011). Regional unemployment and norm-induced effects on life satisfaction. *SOEPpapers on Multidisciplinary Panel Data Research* (387).
- Clark, A. E. (2003). Unemployment as a social norm: Psychological evidence from panel data. *Journal of Labor Economics* 21(2), 323–351.
- Clark, A. E., P. Frijters, and M. A. Shields (2008). Relative income, happiness, and utility: An explanation for the easterlin paradox and other puzzles. *Journal of Economic Literature* 46(1), 95–144.
- Clark, A. E., A. Knabe, and S. Rätzel (2009). Unemployment as a social norm in germany. *Schmollers Jahrbuch* 129(2), 251–260.
- Clark, A. E. and A. J. Oswald (1994). Unhappiness and unemployment. The Economic Journal 104(424), 648–659.
- Diener, E., M. Diener, and C. Diener (1995). Factors predicting the subjective well-being of nations. *Journal of Personality and Social Pschology* 69(5), 851–864.
- Diener, E. and H.-D. Klingelmann (2000). *Culture and subjective well-being*, Chapter Genes, Culture, Democracy, and Happiness, pp. 165–183. MIT Press.
- Diener, E. and W. Tov (2012). *Handbook of Social Indicators and Quality of Life Research*, Chapter National Accounts of Well-Being, pp. 137–157. Springer.
- DiTella, R., R. J. MacCulloch, and A. J. Oswald (2001). Preferences over inflation and unemployment: Evidence from survey of happiness. *The American Economic Review 91*(1), 335–341.

- DiTella, R., R. J. MacCulloch, and A. J. Oswald (2003). The macroeconomics of happiness. *The Review of Economics and Statistics* 85(4), 809–827.
- Easterlin, R. A. (1974). Does economic growth improve the human a lot? some empirical evidence. In P. A. David and M. W. Reder (Eds.), *Nations and House-holds in Economic Growth: Essays in Honor of Moses Abramowitz*, pp. 89–125. Academic Press.
- Easterlin, R. A. (2005). Diminishing marginal utility of income? caveat emptor. *Sical Indicators Research 70*(3), 243–255.
- Easterlin, R. A. and A. C. Plagnol (2008). Life satisfaction and economic conditions in east and west germany pre- and post-unification. *Journal of Economic Behavior and Organization* 68(3-4), 433–444.
- Frey, B. S. and A. Stutzer (2002). *Happiness and Economics*. Princeton University Press.
- Fuchs-Schündeln, N., D. Krueger, and M. Sommer (2010). Inequality trends for germany in the last two decades: A tale of two countries. *Review of Economic Dynamics* 13(1), 103–132.
- Gerlach, K. and G. Stephan (1996). A paper on unhappiness and unemployment in germany. *Economic Letters* 52(3), 325–330.
- Helliwell, J. F. (2003). How's life? combining individual and national variables to explain subjective well-being. *Economic Modelling 20*(2), 331–360.
- Humpert, S. (2010). Machen kinder doch glücklich? SOEPpapers on Multidisciplinary Panel Data Research 301.
- Kahneman, D. and A. B. Krueger (2006). Developments in the measurement of subjective well-being. *Journal of Economic Perspectives* 20(1), 3–24.
- Kahneman, D., A. B. Krueger, D. Schkade, and N. Schwarz (2004). Toward national well-being accounts. *American Economic Review* 94(2), 429–434.
- Kolmos, J. and P. Salamon (2008). The poverty of growth with interdependent utility functions. *The Journal of Socio-Economics* 37(6), 2242–2247.
- Kristensen, N. and E. Johansson (2008). New evidence on cross-country differences in job satisfaction using anchoring vignettes. *Labour Economics* 15(1), 96–117.
- Lane, R. E. (2000). *The Loss of Happiness in Market Democracies*. Yale ISPS Series. New Haven and London: Yale University Press.
- Layard, R. (2010). Measuring subjective well-being. Science 327, 534-535.
- Layard, R., G. Mayraz, and S. J. Nickell (2008). The marginal utility of income. Journal of Public Economics 92(8-9), 1846–1857.
- Luechinger, S. (2011). Valuing air quality using the life satisfaction approach. *The Economic Journal 110*(536), 482–515.

- Luechinger, S., S. Meier, and A. Stutzer (2010). Why does unemployment hurt the employed? evidence from the life satisfaction gap between the public and the private sector. *Journal of Human Resources* 45(4), 989–1045.
- Maddison, D. and K. Rehdanz (2007). Are regional differences in utility eliminated over time? evidence from germany. *SOEPpapers on Multidisciplinary Panel Data Research* (16).
- Ng, Y.-K. (2003). From preference to happiness: Towards a more complete warfare economics. *Social Choice and Welfare 20*(2), 307–350.
- Oishi, S., S. Kesebir, and E. Diener (2011). Income inequality and happiness. *Psychological Science* 22(9), 1095–1100.
- Oswald, A. J. (1997). Happiness and economic performance. *The Economic Journal* 107(445), 1815–1831.
- Oswald, A. J. and S. Wu (2010). Objective confirmation of subjective measures of human well-being: Evidence from the u.s.a. *Science 327*, 576–579.
- Pugno, M. (2009). The easterlin paradox and the decline of social capital: An integrated explanation. *The Journal of Socio-Economics* 38(4), 590–600.
- Raffelhüschen, B., S. Moog, and J. Vatter (2011). *Glücksatlas Deutschland 2011*, Chapter Wie zufrieden ist Deuschland?, pp. 29–138. Deutsche Post.
- Rousseau, J.-B. G. (2009). Happiness and income inequality. mimeo.
- Sachverständigenrat (2010). Wirtschaftsleistung, lebensqualität und nachhaltigkeit: Ein umfassendes indikatorensystem. Sachverständigenrat zur Begutachtung der gesamtwirtschaftlichen Entwicklung and Conseil d'Analyse économique.
- Sacks, D. W., B. Stevenson, and J. Wolfers (2010). Subjective well-being, income, economic development and growth. *NBER Working Paper* (16441).
- Shields, M. A. and S. W. Price (2005). Exploring the economic and social determinants of psychological wellbeing and perceived social support in england. *Journal of the Royal Statistical Society* 168(3), 513–537.
- Stewart, K. (2005). Dimensions of well-being in eu regions: Do gdp and unemployment tell us all we need to know? *Social Indicators Researchn* 73(2), 221–246.
- Veenhoven, R. and J. Ehrhardt (1995). The cross-national pattern of happiness: Test of predictions implied in three theories of happiness. *Social Indicators Research* 34(1), 33–68.
- Vittersø, J., R. Biswas-Diener, and E. Diener (2005). The divergent meanings of life satisfaction: Item response modeling of the satisfaction with life scale in greenland and norway. *Social Indicators Research* 74(2), 327–348.
- Wagner, G. G., J. R. Frick, and J. Schupp (2007). The german socio-economic panel study (soep) scope, evolution and enhancements. *Schmollers Jahrbuch* 127(1), 139–169.

- Wierzbicka, A. (2005). Empirical universals of language as a basis for the study of other human universals and as a tool for exploring cross-cultural differences. *Ethos* 33(2), 256–291.
- Winkelmann, R. and L. Winkelmann (1995). Happiness and unemployment: a panel data analysis for germany. *Konjunkturpolitik* 41(4), 293–307.

Seit 2005 erschienene Beiträge

No. 1	Christian Hagist / Norbert Klusen / Andreas Plate / Bernd Raffelhüschen Social Health Insurance – the major driver of unsustainable fiscal policy?				
No. 2	Stefan Fetzer / Bernd Raffelhüschen / Lara Slawik Wie viel Gesundheit wollen wir uns eigentlich leisten?				
No. 3	Oliver Ehrentraut /Matthias Heidler /Bernd Raffelhüschen En route to sustainability: history, status quo, and future reforms of the German public pension scheme?				
No. 4	Jasmin Häcker/ Bernd Raffelhüschen Die Interne Rendite der Gesetzlichen Pflegeversicherung				
No. 5	Jasmin Häcker / Bernd Raffelhüschen Internal Rates of Return of the German Statutory Long-Term Care Insurance (Englische Fassung von Diskussionsbeitrag No. 4)				
No. 6	Matthias Heidler / Bernd Raffelhüschen How risky is the German Pension System? The Volatility of the Internal Rates of Return				
No. 7	Laurence J. Kotlikoff/ Christian Hagist Who's going broke? Comparing Growth in Healthcare Costs in Ten OECD Countries				
No. 8	Jasmin Häcker Dynamisierung der Pflegeleistungen: Vergangenheit – Gegenwart – Zukunft				
No. 9	Dirk Mevis / Olaf Weddige Gefahr erkannt – Gefahr gebannt? Nachhaltigkeitsbilanz der 15. Legislaturperiode des deutschen Bundestages 2002-2005				
No. 10	Daniel Besendorfer / Emily Phuong Dang / Bernd Raffelhüschen Die Schulden und Versorgungsverpflichtungen der Länder: Was ist und was kommt				
No. 11	Jasmin Häcker/ Bernd Raffelhüschen Zukünftige Pflege ohne Familie: Konsequenzen des "Heimsog-Effekts"				
No. 12	Christian Hagist/ Bernd Raffelhüschen/ Olaf Weddige Brandmelder der Zukunft – Die Generationenbilanz 2004				
No. 13	Matthias Heidler / Arne Leifels / Bernd Raffelhüschen Heterogenous life expectancy, adverse selection, and retirement behavior				
No. 14	Pascal Krimmer / Bernd Raffelhüschen Grundsicherung in Deutschland - Analyse und Reformbedarf				
No. 15	Ulrich Benz / Christian Hagist Konjunktur und Generationenbilanz – eine Analyse anhand des HP-Filters				
No. 16	Jasmin Häcker/ Birgit König / Bernd Raffelhüschen / Matthias Wernicke / Jürgen Wettke Effizienzreserven in der stationären Pflege in Deutschland: Versuch einer Quantifizierung und Impli- kationen für die Reform der Gesetzlichen Pflegeversicherung				
No. 17	Christian Hagist/ Matthias Heidler / Bernd Raffelhüschen / Jörg Schoder Brandmelder der Zukunft – Die Generationenbilanz Update 2007: Demografie trifft Konjunktur				
No. 18	Lukas Mangelsdorff Die Geldsteuer: Vorschlag für eine radikal einfache Steuer				
No. 19	Jasmin Häcker / Tobias Hackmann / Stefan Moog Demenzkranke und Pflegebedürftige in der Sozialen Pflegeversicherung – Ein intertemporaler Kos- tenvergleich				
No. 20	Oliver Ehrentraut/ Matthias Heidler Demografisches Risiko für die Staatsfinanzen? – Koordinierte Bevölkerungsvorsausberechnungen im Vergleich				

No. 21	Oliver Ehrentraut / Matthias Heidler Zur Nachhaltigkeit der GRV – Status quo, Potenziale und Risiken
No. 22	Ulrich Benz/ Christian Hagist Konjunktur und Generationenbilanz – eine Analyse anhand des HP-Filters
No. 23	Ulrich Benz/ Christian Hagist Technischer Anhang zu "Konjunktur und Generationenbilanz – eine Analyse anhand des HP-Filters"
No. 24	Veronika Deeg/ Christian Hagist The Fiscal Outlook in Austria – An Evaluation with Generational Accounts
No. 25	Oliver Ehrentraut/ Bernd Raffelhüschen Demografischer Wandel und Betriebsrenten – Zur Berücksichtigung der Langlebigkeit bei der Anpas- sung von Direktzusagen
No. 26	Tobias Hackmann/ Stefan Moog Älter gleich kränker? Auswirkungen des Zugewinns an Lebenserwartung auf die Pflegewahrschein- lichkeit
No. 27	Klaus Kaier / Christian Hagist / Uwe Frank / Elisabeth Meyer Antimicrobial drug use, alcohol-based hand disinfection and the burden of methicillin-resistant Sta- phylococcus aureus – A time series approach at a German University Medical Center
No. 28	Jasmin Häcker/ Tobias Hackmann/ Thorsten Henne Sozialgesetzgebung und Beihilfeverordnungen: Ein Leistungsvergleich von Versicherten der Sozialen Pflegeversicherung und Beamten im Pflegefall
No. 29	Stefan Moog MacSim: Ein Simulationsmodell zur Analyse der gesamtwirtschaftlichen Auswirkungen der demogra- fischen Entwicklung in Deutschland
No. 30	Christian Hagist / Stefan Moog / Bernd Raffelhüschen Ehrbarer Staat? Die Generationenbilanz – Update 2008: Migration und Nachhaltigkeit
No. 31	Klaus Kaier / Uwe Frank / Christian Hagist / Elisabeth Meyer The impact of antimicrobial drug consumption and alcohol-based hand rub use on the emergence and spread of extended-spectrum β-lactamase (ESBL)-producing strains – A time series analysis
No. 32	Friedrich Fichtner / Christian Hagist Oil and Intergenerational Redistribution – The case of Norway
No. 33	Tobias Hackmann/ Stefan Moog Pflege im Spannungsfeld von Angebot und Nachfrage
No. 34	Christian Hagist / Stefan Moog / Bernd Raffelhüschen / Johannes Vatter Ehrbare Staaten? Die Ergebnisse der Generationenbilanzierung im internationalen Vergleich
No. 35	Christian Hagist / Johannes Vatter Measuring Fiscal Sustainability on the Municipal Level: A German Case Study
No. 36	Uwe Frank / Klaus Kaier Dynamics between antibiotic drug use and resistance – An economic approach
No. 37	Matthias Heidler/ Christoph Müller / Olaf Weddige Measuring accrued-to-date liabilities of public pension schemes – method, data and limitations
No. 38	Stefan Moog / Bernd Raffelhüschen Ehrbarer Staat? Die Generationenbilanz – Update 2009: Wirtschaftskrise trifft Tragfähigkeit
No. 39	Christoph Müller / Bernd Raffelhüschen / Olaf Weddige Measuring pension liabilities and the perspective of sustainability: the case of the reformed German statutory pension scheme

No. 40	Tobias Hackmann Arbeitsmarkt Pflege: Bestimmung der künftigen Altenpflegekräfte unter Berücksichtigung der Berufsverweildauer
No. 41	Astrid L. Grasdal /Jasmin Häcker / Christian Hagist / Erling Vårdal Who pays for whom? Redistributive Effects of the Norwegian Pension Reform
No. 42	Christian Hagist/ Tobias Benz Der Rücklagenbedarf der Versorgungsausgaben in Baden-Württemberg – Projektion und Reform- optionen
No. 43	Jasmin Häcker /Tobias Hackmann LOS(T) in Long-term Care: Empirical Evidence from German Data 2000-2009
No. 44	Stefan Moog /Christoph Müller / Bernd Raffelhüschen Ehrbare Staaten? Die deutsche Generationenbilanz im internationalen Vergleich: Wie gut ist Deutschland auf die demografische Herausforderung vorbereitet
No. 45	Felix Schnurr /Johannes Vatter /Frank Weinmann Zum Kostenanstieg im Gesundheitswesen 2009: Wie teuer waren die Reformen?
No. 46	Jasmin Häcker /Tobias Hackmann / Bernd Raffelhüschen Pflegereform 2010: Karenzzeiten in der Sozialen Pflegeversicherung
No. 47	Janusz Jabłonowski /Christoph Müller /Bernd Raffelhüschen A fiscal outlook for Poland using Generational Accounts
No. 48	Klaus Kaier The impact of pricing and patent expiration on the demand for pharmaceuticals: an examination of the use of broad-spectrum antimicrobials
No. 49	Christian Hagist /Bernd Raffelhüschen /Alf Erling Risa /Erling Vårdal Long-Term Fiscal Effects of Public Pension Reform in Norway – A Generational Accounting Analysis
No. 50	Johannes Vatter Well-Being in Germany: What Explains the Regional Variation?

Forschungszentrum Generationenverträge

Albert-Ludwigs-Universität Freiburg Bertoldstraße 17 79098 Freiburg Fon 0761 . 203 23 54 Fax 0761 . 203 22 90 www.generationenvertraege.de info@generationenvertraege.de